Author’s response to reviews

Title: Urban Health in Africa: A critical global public health priority

Authors:

Jo Vearey (jovearey@gmail.com)
Isaac Luginaah (iluginaa@uwo.ca)
Ng"weina Magitta (ngweina.magitta@gmail.com)
Dativa Shilla (sopdatty@yahoo.com)
Tolullah Oni (tolullah.oni@mrc-epid.cam.ac.uk; Tolu.Oni@mrc-epid.cam.ac.uk)

Version: 3 Date: 11 Mar 2019

Author’s response to reviews:

Thank you for your comments.

We have addressed the comments as follows:

1. Correspondance Article. The abstract should briefly summarize the aim, findings or purpose of the article. An introduction and conclusion heading is not required.

Response: we have edited the abstract accordingly

2. Competing Interest: As a member of the editorial board (Associate Editor) of this journal, in order to ensure transparency, please declare this in the Competing Interests section of the Declarations.

Response: I am NOT a member of the editorial board. On review of the editorial board list on your website (https://bmcpublichealth.biomedcentral.com/about/editorial-board), I am also not aware of involvement of any of the co-authors.

3. Consent For Publication: The current statement is incorrect, please change this to 'not applicable'.

Response: We have edited accordingly.
4. Authors Contribution. Please include a statement in the Authors’ contributions section to the effect that all authors have read and approved the manuscript, and ensure that this is the case (currently stated in Consent For Publication).

Response: We have edited accordingly.

5. Funding. In the section 'Funding', please also describe the role of the funding body/bodies in the design of the study and collection, analysis, and interpretation of data and in writing the manuscript.

Response: We have edited accordingly.

6. Clean Manuscript

Response: a clean manuscript is attached